Author’s response to reviews

Title: Use of local anaesthetics and adjuncts for spinal and epidural anaesthesia and analgesia at German and Austrian University Hospitals: An online survey to assess current standard practice

Authors:

Bianca M Wahlen (wahlen_b@klinik.uni-wuerzburg.de)
Norbert Roewer (Anaesthesie-Direktion@klinik.uni-wuerzburg.de)
Metterlein Thomas (Metterlein_T@klinik.uni-wuerzburg.de)
Peter Kranke (Kranke_P@klinik.uni-wuerzburg.de)

Version: 3 Date: 2 August 2009

Author’s response to reviews:

Dear Dr. Alam

1. "A sample title" has been deleted

2. Information on how this questionnaire was developed have been added in "methods".

3. The online web page has been reactivated, reconstructed and translated into English

B.M.Wahlen